\input{ThisWeek.tex}
\newcommand{\duedate}{Thursday, September 11}
\input{../TemplateWeek/CheckHomeWork.tex}
\input{Exercises.tex}

%\booltrue{teacher}

\begin{enumerate}
\item The $y$ value at the maximum is $0.52913\ldots$
\item You'll get two $x$ values reaching the minimum and two $x$ values reaching the maximum.
\item Just make sure to check that it has the necessary properties; it should be
  \begin{itemize}
  \item Defined on $[-3,3]$,
  \item At $x=-2$ there should be a local minimum
  \item The point at $x=-1$ should be lower than the point at $x=-2$
  \item Indeed, $x=-1$ should be the absolute minimum
  \item There should be a local maximum at $x=2$
  \item The point at $x=1$ should be higher than the point at $x=2$
  \item Indeed, $x=1$ should be an absolute maximum
  \end{itemize}
\end{enumerate}

\end{document}

